Author's response to reviews

Title: Effects of propofol versus thiopental on Apgar scores in newborns and peri-operative outcomes of women undergoing emergency cesarean section: a randomized clinical trial

Authors:

Janat Tumukunde Dr (janattumukunde@gmail.com)
Dlamini D Lomangisi Dr (ngisi2@yahoo.com)
Ocen Davidson Dr (d4davidson@yahoo.co.uk)
Kintu Andrew Dr (kintuug2012@gmail.com)
Ejoku Joseph Dr (ejokuj@gmail.com)
Arthur Kwizera Dr (akwizera@chs.mak.ac.ug)

Version: 7
Date: 5 April 2015

Author's response to reviews: see over
Dear Madam,

Re: Manuscript reference No. MS:9142438631409209

Thank you for accepting our manuscript

Please find attached a revised version of our manuscript “Effects of propofol versus thiopental on Apgar scores in newborns and peri-operative outcomes of women undergoing emergency cesarean section: a randomized clinical trial” with the author’s contribution section and statement confirming that all authors read and approved the final manuscript. Line 288-291

Below is a link to the registered trial.

http://www.pactr.org/ATMWeb/appmanager/atm/atmregistry?_nfpb=true&_pageLabel=atm_portal_page_mytrials

Yours sincerely
Dr Janat Tumukunde